Author's response to reviews

Title: Pirfenidone as salvage treatment for refractory bleomycin-induced lung injury: a case report of seminoma

Authors:
Koji Sakamoto (sakakoji@med.nagoya-u.ac.jp)
Satoru Ito (itori@med.nagoya-u.ac.jp)
Naozumi Hashimoto (hashinao@med.nagoya-u.ac.jp)
Yoshinori Hasegawa (yhasega@med.nagoya-u.ac.jp)

Version: 3 Date: 20 Jul 2017

Author's response to reviews:

Dear Editors of BMC Cancer

Here are the point-by-point reply for your valuable comments.

Q1. Thank you for providing an "Availability of data and materials". However, please amend this to "all data generated or analysed during this study are included in this published article [and its supplementary information files]".

A1. We amended the section of "Availability of data and materials" according to the Editor’s comment. Thank you for the kind consideration.